Author’s response to reviews

Title: Low birth weight AT TERM and the presence of fine particulate matter and carbon monoxide in the Brazilian Amazon: a population-based retrospective cohort study.

Authors:

Ageo M C Silva (ageoms@hotmail.com)
Gisele P Moi (gisele_pedroso_moi@gmail.com)
Inês E Mattos (imattos@ensp.fiocruz.br)
Sandra S Hacon (sandrahacon@gmail.com)

Version: 9 Date: 5 August 2014

Author’s response to reviews:

Subject: MS: 3526024101143988
Low birth weight and presence of fine particulate matter and carbon monoxide in the Brazilian Amazon: a population-based retrospective cohort study.
Ageo M C Silva, Gisele P Moi, Inês E Mattos and Sandra S Hacon

To Ms Janelyn Ann Cruz

Dear Ms. Cruz,

Please, see below the information need to complete the submission process.

1- The abstract was replaced
2- line 160 (STROBE).
3. Lines 112,131 and 132 were complemented.
4. The tables are in the end of the references, as suggested.